Author's response to reviews

Title: Community resistance to a peer education programme in Zimbabwe

Authors:

- Catherine Campbell (c.campbell@lse.ac.uk)
- Kerry Scott (keescott@jhsphs.edu)
- Zivai Mupambireyi (zietawana@yahoo.co.uk)
- Mercy Nhamo (mnhamo@googlemail.com)
- Constance Nyamukapa (c.nyamukapa03@imperial.ac.uk)
- Morten Skovdal (m.skovdal@gmail.com)
- Simon Gregson (Sajgregson@aol.com)

Version: 2 Date: 7 April 2014

Author's response to reviews:

Line numbers + page numbers have been added. Apologies for this oversight.